Author's response to reviews

Title: Interrater and intrarater reliability of photoplethysmography for measuring toe blood pressure and toe-brachial index in people with diabetes mellitus

Authors:

Christopher D Scanlon (Christopher.Scanlon@swahs.health.nsw.gov.au)
Kris E Park (ParkK@wahs.nsw.gov.au)
David Mapletoft (MapletD@wahs.nsw.gov.au)
Lindy Begg (lindybegg@bigpond.com.au)
Joshua Burns (JoshuaB2@chw.edu.au)

Version: 4 Date: 16 May 2012

Author's response to reviews: see over
15th May 2012

Prof Hylton Menz
Editor-in-Chief
Journal of Foot and Ankle Research

Dear Prof Menz,

Ms. Ref. No.: 7225919845550872

Title: Interrater and intrarater reliability of photoplethysmography for measuring toe blood pressure and toe-brachial index in people with diabetes mellitus

We thank the editor and reviewers for their thoughtful comments concerning our manuscript and for considering our study for publication in Journal of Foot and Ankle Research. We have carefully addressed the issues raised by the editorial team and reviewers and would like to take the opportunity to resubmit our revised manuscript.

We have provided our revised manuscript and have addressed the concerns.

As a result, we now believe we have met the requirements and believe this revised manuscript and will be informative to the readership of Journal of Foot and Ankle Research.

Editors’ comments

1. JRAR uses standard English – Amended all unnecessary capitalisation including title
2. Remove author’s qualifications – Author’s qualifications removed

Yours sincerely,

Chris Scanlon and Associate Professor Joshua Burns, Ph.D.
on behalf of the co-authors Kris Park, David Mapletoft, Lindy Begg